Author's response to reviews

Title: Influences of laparoscopic-assisted gastrectomy and open gastrectomy on serum interleukin-6 levels in patients with gastric cancer among Asian populations: a systematic review

Authors:

Zhen Bo Shu (shuzhenbo0722@163.com)
Hai Ping Cao (caohaiping0722@163.com)
Yong Chao Li (liyongchao0722@163.com)
Li Bo Sun (sunlibo0722@163.com)

Version: 6 Date: 18 March 2015

Author's response to reviews: see over
Author's response to reviews

Title: Influences of laparoscopic-assisted gastrectomy and open gastrectomy on serum interleukin-6 levels in patients with gastric cancer among Asian populations: a systematic review

Authors:

Zhen-Bo Shu (shuzhenbo0722@163.com)
Hai-Ping Cao (caohaiping0722@163.com)
Yong-Chao Li (liyongchao0722@163.com)
Li-Bo Sun (sunlibo0722@163.com)

Version: 2   Date: 24 January 2015

Author's response to reviews: see over
Reviewer's report

Title: Influences of laparoscopic-assisted gastrectomy and open gastrectomy on serum interleukin-6 levels in patients with gastric cancer among Asian populations: a systematic review

Version: 3    Date: 13 January 2015

Reviewer's report:

Please include in the title of the manuscript and also in the submission system that "this was a systematic review".

Response: Thank you. We have revised the title of the manuscript and also in the submission system that “Influences of laparoscopic-assisted gastrectomy and open gastrectomy on serum interleukin-6 levels in patients with gastric cancer among Asian populations: a systematic review".